Author's response to reviews

Title: Prevention, control, and elimination of neglected diseases in the Americas: Pathways to integrated, inter-programmatic, inter-sectoral action for health and development

Authors:

John C Holveck (johnholveck@yahoo.com)
John P Ehrenberg (ehrenbeij@paho.org)
Steven K Ault (aultstev@paho.org)
Rocio Rojas (rorojas@ecu.ops-oms.org)
Javier Vasquez (vasquezj@paho.org)
Maria Teresa Cerqueira (cerqueim@paho.org)
Josefa Ippolito-Shepherd (ippolij@paho.org)
Miguel A Genovese (genoves@paho.org)
Mirta Roses Periago (roesesper@paho.org)

Version: 4 Date: 12 January 2007

Author's response to reviews: see over
SUBJECT: MS: 9535718789764347

Prevention, control, and elimination of neglected diseases in the Americas: Pathways to integrated, inter-programmatic, inter-sectoral action for health and development

John C Holveck, John P Ehrenberg, Steven K Ault, Rocio Rojas, Javier Vasquez, Maria Teresa Cerqueira, Josefa Ippolito-Shepherd, Miguel A Genovese and Mirta Roses Periago

BMC Public Health

12 January 2007

We believe we have responded to all of the formatting changes requested. Please also note that for clarification we have made minor changes to the section on authors’ contributions.

Point-by-point description of formatting changes:

1. We have removed the text ‘Debate’ from the title page.
2. Author’s affiliations have been linked using superscript numbers as directed.
3. Author’s email addresses have been listed after their affiliation.
4. Author’s job titles have been removed from the title page.
5. The corresponding author (Steven K. Ault) has been indicated using an asterisk mark. His email address is included after his affiliation.
6. The text in the boxes has been incorporated into the main text of the manuscript. One of the text boxes has been converted to a table and is now presented at the end of the manuscript, after the title of the figure.
7. The authors’ have all reviewed and approved the final manuscript. This is indicated under the section on Author’s Contributions.
8. The text REFERENCES has been changed to References as directed.
9. The figure title has been removed from the image file as directed.
10. The background section heading has been changed as directed.